For reprint orders, please contact: reprints@futuremedicine.com

Enhanced metastatic risk assessment in cutaneous squamous cell carcinoma with the 40-gene expression profile test

Sherrif F Ibrahim^{*,1,2}, Julia M Kasprzak³, Mary A Hall⁴, Alison L Fitzgerald⁴, Jennifer J

Siegel⁴, Sarah J Kurley⁴, Kyle R Covington⁴, Matthew S Goldberg^{4,5}, Aaron S Farberg⁶, Shannon C Trotter⁷, Kenneth Reed⁸, David G Brodland⁹, Shlomo A Koyfman¹⁰, Ally-Khan

Somani¹¹, Sarah T Arron¹² & Ashley Wysong¹³

¹Rochester Dermatologic Surgery, Victor, NY 14564, USA

³Department of Dermatology, Medical College of Wisconsin, Milwaukee, WI 53226, USA

⁶Department of Dermatology, Baylor University Medical Center, Dallas, TX 75246, USA

⁸DermASAP, Quincy, MA 02169, USA

¹¹Department of Dermatology, Indiana University School of Medicine, Indianapolis, IN 46202, USA

¹²Sarah Arron, P.C., San Mateo, CA 94115, USA

¹³Department of Dermatology, University of Nebraska Medical Center, Omaha, NE 68198, USA

*Author for correspondence: sherrif_ibrahim@urmc.rochester.edu

Aim: To clinically validate the 40-gene expression profile (40-GEP) test for cutaneous squamous cell carcinoma patients and evaluate coupling the test with individual clinicopathologic risk factor-based assessment methods. Patients & methods: In a 33-site study, primary tumors with known patient outcomes were assessed under clinical testing conditions (n = 420). The 40-GEP results were integrated with clinicopathologic risk factors. Kaplan–Meier and Cox regression analyses were performed for metastasis. Results: The 40-GEP test demonstrated significant prognostic value. Risk classification was improved via integration of 40-GEP results with clinicopathologic risk factor-based assessment, with metastasis rates near the general cutaneous squamous cell carcinoma population for Class 1 and \geq 50% for Class 2B. Conclusion: Combining molecular profiling with clinicopathologic risk factor assessment enhances stratification of cutaneous squamous cell carcinoma patients and may inform decision-making for risk-appropriate management strategies.

Plain language summary: Cutaneous squamous cell carcinoma is a common skin cancer, with approximately 2 million cases diagnosed each year in the USA. Because substantial numbers of patients experience metastasis, which can result in death, accurate metastatic risk assessment is important. Clinicians use clinicopathologic factors to determine risk for disease progression. However, traditional methods miss pinpointing many patients who experience metastasis and sometimes categorize patients as at risk who do not develop metastasis, indicating that additional tools are needed. A molecular test, the 40-gene expression profile (40-GEP), was developed to predict metastatic risk based on the biology of the tumor. This study demonstrates that the 40-GEP, either as an independent tool or together with traditional methods, accurately identifies patients' risk of metastasis. Using the 40-GEP could improve patient management to improve patient outcomes.

First draft submitted: 7 October 2021; Accepted for publication: 8 November 2021; Published online: 25 November 2021

Keywords: clinicopathologic risk factors • cutaneous squamous cell carcinoma • gene expression profile • informed decision-making • metastasis/metastatic risk



Future

²Department of Dermatology, University of Rochester Medical Center, Rochester, NY 14620, USA

⁴Castle Biosciences, Inc., Friendswood, TX 77546, USA

⁵Department of Dermatology, Icahn School of Medicine at Mount Sinai, New York, NY 10025, USA

⁷Oakview Dermatology, Springfield, OH 45504, USA

⁹Zitelli & Brodland, P.C., Pittsburgh, PA 15232, USA

¹⁰Department of Radiation Oncology, Cleveland Clinic, Cleveland, OH 44106, USA

An estimated 2–6% of cutaneous squamous cell carcinoma (cSCC) patients develop regional or distant metastasis, and approximately 2% die from the disease annually in the USA. [1–6] Although the fatality rate is low, the incidence of cSCC is high (\sim 2 million diagnosed cases/year) and continues to grow, resulting in a substantial number of patients with poor outcomes [7–11]. As the distribution of nonmelanoma skin cancer is shifting from a historical 1:4 toward a 1:1 ratio of cSCC to basal cell carcinoma, the estimated mortality rate of cSCC will likely surpass that for melanoma [8,9].

Development of metastatic disease has a profound impact on cSCC patient survival, as most cSCC deaths are due to progressive locoregional disease (e.g., regional lymph node metastasis or in-transit metastasis) [1,12]. This underscores the need for effective identification of patients at risk of metastasis. While the 5-year disease-specific survival rate is >90% for localized disease [6,13], this rate drops to 50–83% and below 40% for patients with regional and distant metastases, respectively. Thus, although interventions are available for treatment of locally advanced cSCC [14,15], accurate and early identification of cSCC tumors with higher metastatic potential is essential for optimizing patient management to reduce and prevent metastasis and death from an otherwise curable disease. Once metastasis is detected, more intense treatment is often warranted [8].

Many patients with cSCC are broadly classified as having high-risk disease based on clinicopathologic factors associated with increased likelihood of poor outcomes [6,8]. The most current National Comprehensive Cancer Network (NCCN) guidelines categorize a patient as being high risk or very high risk by the presence of specific high-risk factors, indicating that the patient is a candidate for more intensive management [6]. Tumors with diameter ≥ 2 cm have demonstrated two- and threefold greater risk for recurrence and metastasis, respectively, relative to smaller tumors [8,16]. Likewise, tumors invading beyond subcutaneous fat, with perineural invasion (PNI) of large-caliber nerves, or with poor histologic differentiation have been linked to a twofold to 23-fold increased risk for recurrence and/or metastasis in univariate analyses [8,16]. At the patient level, immunosuppressed individuals are at greater risk for developing cSCC and often present with more aggressive tumors [8,17]. While these and other factors are used to stratify patient risk, low accuracy, histopathologic discordance and a lack of standardized reporting limit the clinical utility of this clinicopathologic factor-based approach.

Although methods for cSCC risk stratification continue to be refined, a single universally accepted system has not yet been adopted [18]. Current tumor (T) staging systems include the American Joint Committee on Cancer (AJCC) Cancer Staging Manual, Eighth Edition (AJCC8) [19] and the Brigham and Women's Hospital (BWH) system [20]. AJCC8 has improved upon previous editions for cSCC, albeit specifically noted for use in head and neck tumors only, and the BWH tumor classification system has been reported to provide better prognostication of localized cSCC relative to AJCC8 [18,21]. Both systems utilize clinicopathologic factors of the primary tumor to categorize risk into T stages. For stratifying metastatic risk in cSCC, these systems have demonstrated positive predictive value (PPV) ranges of 14-17 (AJCC8) and 24-38% (BWH) [20-24]. Staging systems do not take into account all risk factors known to be associated with metastasis, and discordance in assessment and reporting has been reported [3,20,23,25,26]. Published reports demonstrate that approximately 30% of cases having metastatic outcomes are misclassified as low-risk T stage (e.g., BWH T1/T2a), and approximately 75% of those classified as high-risk T stage (e.g., BWH T2b/T3) do not develop metastasis [20,21,24,27-30]. This suggests that, while staging has improved, there remains a need for refinement of risk assessment in high-risk cSCC. Given these clinical limitations, physicians may rely on professional experience and institution-specific approaches, rather than staging systems, to drive patient management decisions [31]. Despite ongoing efforts to improve risk assessment [32], a standardized and accurate stratification system remains a clinically unmet need in the management of patients with high-risk cSCC.

We previously reported validation of a gene expression profiling-based algorithm (the 40-gene expression profile [40-GEP]) that stratifies cSCC tumor risk of metastasis [23]. In this study we demonstrate the clinical validity of the 40-GEP test for classifying low (Class 1), moderate (Class 2A) and high (Class 2B) metastatic risk in high-risk cSCC, along with its ability to further stratify risk within the newly defined NCCN risk groups; we also combine 40-GEP test results with clinicopathologic factor-based risk assessment. In an expanded cohort of cSCC patients (n = 420) with high-risk factors, we show significant prognostic value of the 40-GEP test performed using clinical laboratory-developed and implemented standard operating procedures (SOPs) on primary tumor specimens from biopsy or surgical excision. Combining novel molecular prognostication with clinicopathologic risk assessment

approaches can complement current systems and support overall metastatic risk stratification. This, in turn, could facilitate management decisions for high-risk cSCC patients.

Patients & methods

Study cohort

Using an ongoing, institutional review board-approved protocol, formalin-fixed paraffin-embedded (FFPE) samples from primary cSCC lesions (either biopsy or wide local excision specimens) and corresponding clinicopathologic and outcomes data were collected (Supplementary Figure 1). The study design targeted enrollment of patients with \geq 1 high-risk feature as defined by NCCN guidelines or by AJCC or BWH staging greater than T1. The primary end point of the study was 3-year metastasis-free survival (MFS). Study inclusion/exclusion criteria were as previously reported [23]. Cases had a documented regional or distant metastasis, or documented follow-up of at least 3 years post-diagnosis of the primary tumor without a metastatic event, the time by which almost all metastases develop in cSCC [13,33]. Cases with prior history of cSCC, cutaneous basal cell carcinoma or melanoma *in situ* were allowed if deemed cured by the investigator. All cases, when available, underwent review of biopsy pathology reports, definitive surgery reports and medical records. Clinicopathologic factors were deemed positive/present if identified during any review step, resulting in comprehensive staging.

As shown in Supplementary Figure 1, 420 cases (307 previously assessed prior to implementation of clinical SOPs [23] and 113 novel) made up the clinical validation cohort, from which tissue samples were independently reviewed by a board-certified dermatopathologist for tumor content and high-risk factors [6,34,35]. Cases with synchronous tumors were not included in the 420-case clinical validation cohort, as they were reserved for future intrapatient heterogeneity analysis. Also, cases lacking sufficient material to be run under clinical SOPs or not meeting clinical testing criteria were excluded from the study. Seven risk factors were assessed to generate an overall risk factor count: tumor size and location, immunosuppression, PNI, depth of invasion, degree of differentiation, histologic subtype and lymphovascular invasion (Supplementary Table 1). Also, based on comprehensive review (as described above), each case was staged according to AJCC8 and BWH tumor classification systems.

Gene expression analysis

All samples were tested under 40-GEP clinical SOPs in a central College of American Pathologists-accredited laboratory, with personnel blinded to patient outcomes. Briefly, FFPE tumor tissue was macrodissected, processed for real-time PCR and run in triplicate, as previously described [23]. Duplicate sample runs were used to generate clinical 40-GEP class scores.

Statistics

Statistical analyses were performed in R (v3.6.3) as previously described [23]. MFS analyses were performed using Kaplan–Meier methods and log-rank test. Univariate and multivariate Cox regression analyses were performed using standard methods, and assumptions of proportional hazards were confirmed using the zph test of the fitted model for Cox regression, as previously described [23]. Binary T stages (consistent with previous reports) [20–24] as well as individual T stages, determined by comprehensive review as described above, were included in the analyses. The McNemar test was used for comparing accuracy metrics [20].

Results

Cohort characteristics & risk assessment by molecular prognostication

To validate the 40-GEP test for use in the clinical setting (when at least one high-risk factor is present at time of biopsy or surgical excision), FFPE samples from primary cSCC tumors meeting protocol and clinical testing inclusion criteria were assessed using the previously published 40-GEP algorithm [23] and unreported clinical SOPs. The clinical validation cohort (n = 420) included 63 cases with regional and/or distant metastases, and 357 cases without an event and with \geq 3 years of follow-up; cohort characteristics are shown in Table 1. All 63 metastatic cases had regional disease (which included satellite, in-transit, nodal or parotid metastases), 17 of which also developed distant metastases. Median time to metastasis was 0.9 years (95th percentile: 2.7 years). The following characteristics had significantly different rates for metastatic cases: male sex, head and neck location, tumor diameter, tumor thickness, poor differentiation, PNI, invasion beyond subcutaneous fat, cases undergoing Mohs micrographic surgery and cases deemed very high risk by NCCN (Table 1).

The 40-GEP test classified patients based on risk for regional and/or distant metastasis (Figure 1A). Of the 420

Table 1. Demographics and clinical characteristics of the study cohort (n = 420).						
Characteristic	All (n = 420)	Nonmetastatic (n = 357)	Regional/distant metastasis (n = 63)	p-value		
Age (years):median (range)	71 (34–95)	71 (34–95)	70 (44–90)	ns		
Male sex	308 (73.3%)	253 (70.9%)	55 (87.3%)	0.007		
Caucasian	417 (99.3%)	355 (99.4%)	62 (98.4%)	ns		
Immunosuppression [†]	103 (24.5%)	83 (23.2%)	20 (31.7%)	ns		
Located on H&N	278 (66.2%)	224 (62.7%)	54 (85.7%)	0.0002		
– Ear	64 (15.2%)	53 (14.8%)	11 (17.5%)			
– Lip	25 (6.1%)	17 (4.8%)	8 (12.7%)			
– Scalp	56 (13.3%)	40 (11.2%)	16 (25.4%)			
Tumor diameter (cm):mean (SD) ‡	2.01 (±1.86)	1.84 (±1.67)	3.11 (±2.52)	<0.0001		
Tumor thickness (mm):mean (SD) \S	4.34 (±6.45)	3.72 (±6.63)	7.71 (±4.07)	<0.0001		
Poorly differentiated	58 (13.8%)	36 (10.1%)	22 (34.9%)	<0.0001		
PNI present (≥0.1 mm)	7 (1.7%)	5 (1.4%)	2 (3.2%)	<0.0001		
– Present (<0.1 mm)	22 (5.2%)	12 (3.4%)	10 (15.9%)			
– Present (unknown caliber)	24 (5.7%)	17 (4.8%)	7 (11.1%)			
– Not present	367 (87.4%)	323 (90.5%)	44 (69.8%)			
Invasion beyond subcutaneous fat	51 (12.1%)	34 (9.5%)	17 (27.0%)	<0.0001		
Definitive surgery MMS [¶]	333 (79.3%)	291 (81.5%)	42 (66.7%)	0.023		
NCCN v2.2021 high risk	255 (60.7%)	230 (64.4%)	25 (39.7%)	<0.001		
– Very high risk	165 (39.3%)	127 (35.6%)	38 (60.3%)			

Data analyzed using χ -square test or Kruskal–Wallis F test as appropriate for variable type.

[†]86 of 103 immunosuppressed patients were transplant patients.

[‡]Tumor diameter reported (n = 393).

 ${}^{\mbox{S}}$ Tumor thickness reported (n = 123).

[¶]MMS or wide local excision (n = 415), with two cases not having additional surgery beyond biopsy and three cases with unknown definitive surgery.

H&N: Head and neck; MMS: Mohs micrographic surgery; NCCN: National Comprehensive Cancer Network; ns: Not statistically significant; PNI: Perineural invasion; SD: Standard deviation.







1 risk factor

Overall

event rate

4.0%

10.8%

60.0%

8.2%



≥2 risk factors				
3-year MFS (95% CI)	Overall event rate			
91.9% (87.0–97.1%)	9.0%			
75.8% (68.5–83.9%)	25.0%			
50.0% (31.5–79.4%)	50.0%			
81.1% (76.4–86.1%)	19.7%			

Figure 1. Kaplan–Meier analysis of prognostication by 40-gene expression profile results alone and after combining with risk factor counts. (A) Prognostication by 40-GEP results only. (B) Subset with one risk factor. (C) Subset with two or more risk factors. 40-GEP: 40-gene expression profile; MFS: Metastasis-free survival.

3-year MFS (95% CI)

96.0% (92.3-99.9%)

89.2% (82.0-97.1%)

40.0% (13.7-100%)

91.8% (87.7-96.0%)



Figure 2. Kaplan–Meier analysis of prognostication by 40-gene expression profile results in combination with National Comprehensive Cancer Network (v2.2021) risk classification. (A) NCCN high risk. (B) NCCN very high risk. 40-GEP: 40-gene expression profile; MFS: Metastasis-free survival; NCCN: National Comprehensive Cancer Network.

cases included in the study, 212 were identified as Class 1 (low risk), 185 as Class 2A (moderate risk) and 23 as Class 2B (high risk), with metastasis rates of 6.6, 20.0 and 52.2%, respectively. The 3-year MFS rates were 93.9, 80.5 and 47.8% for Class 1, Class 2A & B, respectively (log-rank, p < 0.001).

Combination of the 40-GEP with clinicopathologic factor-based risk assessment

To determine the impact of molecular prognostication on existing risk-assessment strategies, subsets composed of molecular class and risk factors were interrogated by Kaplan–Meier and regression analysis. First, cases were assessed for total count of risk factors, as determined by NCCN risk criteria [6] or Mohs micrographic surgery appropriate use criteria [34,35] (as described in the methods; Supplementary Table 1), and then binned into two subsets: those with one risk factor (n = 171) and those with two or more risk factors (n = 249; Figure 1B & C). A direct relationship between risk factor count and metastasis rate was demonstrated. The metastasis rate for cases with a single risk factor was 8.2%, while it was 19.7% for cases with two or more risk factors (compared with 15.0% for the whole cohort). Incorporating 40-GEP test results identified Class 1 cases with rates of 4.0 and 9.0%, respectively, for one and two or more risk factors (Figure 1B & C). Combining Class 2A results with risk factors identified cases with higher metastasis rates relative to pre-40-GEP testing (10.8 and 25.0% for one and two or more factors, respectively). Regardless of risk factor count, Class 2B metastasis rates were 50–60% (>twofold above the rate for each subset without 40-GEP prognostication), which represent statistically equivalent findings within CIs. These findings were supported by significantly different 3-year MFS rates for Class 1, 2A & B cases in the cohort and corresponding rate changes for each subset (log-rank p < 0.0001; Figure 1).

When criteria from the newly developed NCCN guidelines were applied [6], all 420 patients of the cohort were defined as either high or very high risk (with 3-year MFS rates of 90.2 and 78.2%, respectively; p = 0.0002), confirming the cohort's high-risk designation (Figure 2). The 40-GEP test further stratified risk within each of the NCCN groups (Figure 2A & B). For the high-risk subset, MFS rates were 95.9, 84.3 and 62.5% for Class 1, Class 2A & B, respectively (p = 0.0001); for the very high-risk subset, MFS rates were 89.6, 75.9 and 40.0% for Class 1, Class 2A & B, respectively (p < 0.001).

Table 2. Accuracy metrics for the cutaneous squamous cell carcinoma validation cohort (n = 420) per 40-gene expression profile class and Brigham and Women's Hospital or American Joint Committee on Cancer Cancer Staging Manual, Eighth Edition binary T stage.

Metric	Accuracy (%)					
	40-GEP (Class 2)	40-GEP (Class 2B)	BWH (T2b/T3)	AJCC8 (T3/T4)		
Sensitivity [†]	77.8	19.0	30.2	38.1		
Specificity [‡]	55.5	96.9	89.6	84.3		
PPV	23.6	52.2	33.9	30.0		
NPV	93.4	87.2	87.9	88.5		

[†]p < 0.0001 for a 40-GEP Class 2 result vs BWH T2b/T3 stage or AJCC8 T3/T4 stage; p < 0.02 for a 40-GEP Class 2B result vs AJCC8 T3/T4 stage.

*p < 0.0001 for a 40-GEP Class 2B result vs BWH T2b/T3 stage or AJCC8 T3/T4 stage, and for a 40-GEP Class 2 result vs BWH T2b/T3 stage or AJCC8 T3/T4 stage; based on the McNemar test.

Note: p-values were not estimated for PPV and NPV, as these values are based on prevalence of metastasis.

40-GEP: 40-gene expression profile; AJCC8: American Joint Committee on Cancer Staging Manual, Eighth Edition; BWH: Brigham and Women's Hospital; NPV: Negative predictive value; PPV: Positive predictive value.

Next, after T stages for comprehensively assessed tumors were determined (AJCC8 and BWH), accuracy metrics were calculated using previously published groupings [20–24], and metastatic risk rates within T stage-defined risk groups were stratified by 40-GEP class. When the accuracy metrics of the 40-GEP test were compared with those of AJCC8 and BWH T staging for this cohort, the PPV for a 40-GEP Class 2B result was 52.2% compared with 30.0 and 39.9% for high-stage AJCC8 (T3/T4) and BWH (T2b/T3), respectively, while maintaining a similar negative predictive value (87.2% compared with 88.5 and 87.9%; Table 2). The specificity of a 40-GEP Class 2B result (96.9%) was significantly higher (p < 0.0001) than that of high-stage AJCC8 (84.3%) and BWH (89.6%), demonstrating that a Class 2B result has a significantly reduced likelihood of yielding a false positive regarding metastasis relative to AJCC8 and BWH high-stage classifications. The sensitivity of a 40-GEP Class 2 (Class 2A or 2B) result (77.8%) was significantly higher (p < 0.0001) than that of high-stage AJCC8 (38.1%) and BWH (30.2%). This indicates that a Class 2 result is significantly more likely to successfully identify tumors at moderate to high risk of metastasis (i.e., it demonstrates fewer false negatives) relative to AJCC8 and BWH staging approaches.

Analyses of metastasis rates stratified by 40-GEP class and either binary T stage (a grouping used for clinical decision-making and accuracy metrics) [5,20] or individual T stage (as distinct risk groups) are reported in Figure 3, Supplementary Figure 2 & Supplementary Table 2. Addition of molecular testing results to binary T stage status identified subpopulations ranging from 5.7 to 71.4% (BWH) and from 5.6 to 83.3% (AJCC8), compared with 12.1–33.9% (BWH) and 11.5–30.0% (AJCC8) for binary staging alone, demonstrating that risk assessment can be refined by combining the two approaches. Further, when metastatic rates for individual BWH T stages were assessed per 40-GEP class, a similar trend of increasing rates of metastasis was observed from T1 to T3 within classes and from Class 1 to 2B. Two exceptions were a lower rate for T3 stage within Class 1 & 2B, which also had relatively low numbers of cases (Figure 3, Supplementary Figure 2 & Supplementary Table 2). Metastatic rates for individual AJCC8 T stages were also compared and found to show similar trends (Figure 3, Supplementary Figure 2 & Supplementary Table 2).

To determine the contributions of individual factors to metastatic risk, factors with best-supported evidence for association with metastasis [6,8] and molecular prognostication by the 40-GEP were assessed by Cox regression analyses (Table 3). By univariate analysis, the risk of metastasis for Class 2A & B results was 3.22- and 11.61-fold greater, respectively, than that for Class 1 results (p < 0.001). The presence of poor differentiation, PNI and deep invasion (i.e., beyond subcutaneous fat, depth >6 mm, or Clark level V) were significant risk factors for metastasis (hazard ratios [HRs]: 3.93, 3.28 and 3.11, respectively; p < 0.001). Tumor diameter was also significantly predictive of metastatic risk (HR: 1.15 per cm; p < 0.001). Despite prior support for immunosuppression as a prognostic risk factor [8,36], it was not statistically significant in this cohort. It should be noted that lymphovascular invasion and increased tumor thickness were associated with immunosuppression (data not shown); however, lymphovascular invasion was rare in this cohort. Univariate analysis aligned with NCCN risk grouping, as NCCN very high-risk cases had a 2.54-fold increased risk of metastasis over cases with a high-risk designation (p < 0.001).

When a multivariate model was generated using factors found to be significant in the univariate analysis, 40-GEP results, poor differentiation and deep invasion were independent factors significantly associated with metastatic risk (Table 3). Similar HRs were observed for a Class 2A result, poor differentiation and deep invasion (2.33, 2.29 and





40-GEP: 40-gene expression profile; AJCC8: American Joint Committee on Cancer Cancer Staging Manual, Eighth Edition; BWH: Brigham and Women's Hospital; cSCC: Cutaneous squamous cell carcinoma; T: Tumor.

2.05, respectively; p < 0.05), indicative of independent and additive risk associated with each factor. Overall, a 40-GEP Class 2B result had the greatest independent prognostic value (HR: 6.86; p < 0.001). Multivariate analysis of 40-GEP risk assessment and NCCN risk grouping demonstrated statistical significance of Class 2A & B, with HRs of 2.92 and 9.50, respectively (p < 0.001), while the NCCN very high-risk category had a significant HR of 1.99 (p = 0.009; Table 3).

When univariate analysis was performed for BWH and AJCC8 T stage (binary or individual), HRs were statistically significant for binary high-stage BWH (T2b/T3, 2.55; p = 0.009) and AJCC8 (T3/T4, 2.14; p = 0.022; Table 3), as well as for individual BWH T2b or T3 (3.97 or 4.68; $p \le 0.005$) and AJCC8 T3 or T4 (2.47 or 3.46; $p \le 0.020$; Supplementary Table 3). Also, when compared with BWH or AJCC8 T staging (binary or individual) in univariate or multivariate analyses, a 40-GEP Class 2B result demonstrated more than twofold greater independent prognostic value (p < 0.001; Table 3 & Supplementary Table 3).

Table 3. Univariate and multivariate Cox regression analyses of risk of metastasis per 40-gene expression profile class, common risk factors for poor outcomes, National Comprehensive Cancer Network; risk group and binary Brigham and Women's Hospital and American Joint Committee on Cancer Cancer Staging Manual, Eighth EditionT stages in cutaneous squamous cell carcinoma validation cases.

Risk factor	n	Univariate Cox regression		Multivariate Cox regression [†]	
		Hazard ratio (95% CI)	p-value	Hazard ratio (95% CI)	p-value
40-GEP result					
Class 1	212	1.00 (–)	-	1.00 (–)	-
Class 2A	185	3.22 (1.74–5.95)	<0.001	2.33 (1.20–4.53)	0.013
Class 2B	23	11.61 (5.36–25.15)	<0.001	6.86 (2.73–17.22)	< 0.001
Clinicopathologic risk factor	rs				
Poor differentiation	58	3.93 (2.34–6.60)	<0.001	2.29 (1.21–4.33)	0.011
Perineural invasion [‡]	53	3.28 (1.41–14.36)	<0.001	1.22 (0.58–2.59)	0.601
Deep invasion [§]	72	3.11 (1.86–5.20)	<0.001	2.05 (1.04–4.04)	0.039
Tumor diameter ¶	N/A	1.15 (1.08–1.22)	<0.001	1.07 (0.97–1.17)	0.188
Immunosuppression	103	1.46 (0.86–2.49)	0.161	-	-
40-GEP result					
Class 1	212	1.00 (–)	-	1.00 (–)	-
Class 2A	185	3.22 (1.74–5.95)	<0.001	2.92 (1.57–5.44)	< 0.001
Class 2B	23	11.61 (5.36–25.15)	<0.001	9.50 (4.33–20.9)	< 0.001
NCCN risk group					
High	255	1.00 (–)	-	1.00 (–)	-
Very high	165	2.54 (0.26–3.62)	<0.001	1.99 (1.19–3.33)	0.009
40-GEP result					
Class 1	212	1.00 (–)	-	1.00 (–)	-
Class 2A	185	3.22 (1.74–5.95)	<0.001	2.98 (1.61–5.53)	< 0.001
Class 2B	23	11.61 (5.36–25.15)	<0.001	9.42 (4.28–20.7)	< 0.001
BWH T stage					
T1/T2a	364	1.00 (–)	-	1.00 (–)	-
T2b/T3	56	2.55 (1.26–5.17)	0.009	2.38 (1.38–4.13)	0.002
40-GEP result					
Class 1	212	1.00 (–)	-	1.00 (–)	-
Class 2A	185	3.22 (1.74–5.95)	<0.001	2.97 (1.60–5.51)	< 0.001
Class 2B	23	11.61 (5.36–25.15)	<0.001	11.40 (5.26–24.69)	< 0.001
AJCC8 T stage					
T1/T2	340	1.00 (–)	-	1.00 (–)	-
T3/T4	80	2.14 (1.11–4.09)	0.022	2.69 (1.61–4.48)	< 0.001

[†]n = 393, 54 events, excluding cases without tumor diameter reported.

[‡]Perineural invasion was considered positive regardless of nerve caliber.

 $^{\circ}$ Deep invasion: beyond the subcutaneous fat, depth >6 mm or Clark level V.

¶Tumor diameter: continuous variable per cm.

40-GEP: 40-gene expression profile; AJCC8: American Joint Committee on Cancer Cancer Staging Manual, Eighth Edition; BWH: Brigham and Women's Hospital; cSCC: Cutaneous squamous cell carcinoma; N/A: Not applicable; NCCN: National Comprehensive Cancer Network; T: Tumor.

Discussion

We propose that molecular prognostication, in conjunction with patient and tumor characteristics, increases accuracy of risk assessment for patients with high-risk cSCC. Here we clinically validated the 40-GEP test to identify cSCC tumors at low (Class 1), moderate (Class 2A) and high (Class 2B) risk of metastasis within 3 years of diagnosis, the time by which most metastatic events occur [6,13]. This study follows the previously reported validation of the 40-GEP algorithm for determining metastatic risk [23], validates the test under SOPs implemented for clinical application with primary tumor specimens from biopsy or surgical excision, and demonstrates impactful incorporation with clinicopathologic risk factor-based assessment.

Clinicopathologic risk prediction methods have been improving, in particular with the Eighth Edition updates to AJCC staging and validation of the BWH T-staging system [20-22,24]. However, these systems face the following challenges: lack of standardized reporting; subjective nature of histopathologic assessment; [6] and failure to capture biological risk at the molecular level. For example, differentiation was removed from AJCC8 tumor staging as the definitions of good, moderate and poor differentiation were inconsistent, limiting its clinical application [19]. While tumor depth is a well-accepted risk factor for metastasis, how this variable should be captured/reported and what degree of invasion is considered high risk is debated [37]. Additionally, the rarity of large-nerve PNI [21] (1.7% in this study) may limit its widespread utility for identifying high-risk patients. These caveats likely contribute to the low PPV associated with traditional risk assessment, supporting the need for objective and consistent molecular tools to assess tumor biology.

Molecular biomarker assays and gene expression profiling have demonstrated clinical utility for patient management decision-making in many types of cancer [38–44], although the availability of a clinically validated molecular test to assess metastatic risk in cSCC has been lacking until recently. Descriptive molecular characterization of cSCC has previously identified genes involved in disease progression, and biomarkers have been proposed [45–49]. However, low availability of quality samples has contributed to limitations in the clinical validation and application of many molecular testing strategies for cSCC. The current study demonstrates clinical validation of the 40-GEP test using clinically implemented procedures.

Integration of molecular prognostication into risk assessment can mitigate the limitations of assessment based on clinicopathologic factors alone [20,25,26]. The findings from this study demonstrate that the 40-GEP test can complement clinicopathologic factor-based risk assessment by the use of either individual risk factors or tumor staging. In a multivariate model with commonly utilized high-risk factors, the 40-GEP test provided independent and additive prognostic value. Class 2A & B results were higher and equivalent indicators, respectively, of risk of metastasis relative to other significant factors (poor differentiation and deep invasion; Table 3), indicating that the 40-GEP can augment risk classification of cSCC. Both deep invasion and tumor diameter were significant risk factors in univariate analysis; however, tumor diameter was not significant in multivariate analysis, suggesting that the directionality of growth (i.e., invasive behavior vs surface spread) provides an important distinction. Furthermore, PNI was only statistically significant in univariate analysis, consistent with prior studies showing that PNI loses significance in analysis with other high-risk factors [27]. Also, immunosuppression did not reach statistical significance for association with metastasis, despite the fact that it has been strongly associated with risk for multiple cSCC lesions and poor outcomes [8,17,36]. Interestingly, immunosuppression is not indicated within risk criteria of BWH or AJCC8 tumor staging systems [20]. Relative to the general cSCC patient population, similar metastasis rates have been reported in organ transplant recipients who develop cSCC, although the aggressiveness of disease in immunosuppressed individuals underscores the importance of intensive patient surveillance and management [50].

In this study we demonstrate that the 40-GEP test can enhance clinicopathologic risk factor-based assessment and identify a group of cSCC patients within a high-risk cohort who have metastasis rates similar to those of the general cSCC population (40-GEP Class 1 with one risk factor). With two or more risk factors, the metastasis rate for patients with a Class 1 result was below 10% (\geq 50% lower than the rate for the whole cohort). Likewise, patients having an NCCN high-risk classification and a 40-GEP Class 1 result had metastasis rates similar to those of the general cSCC population; however, the rate was more than twofold higher when 40-GEP risk stratification was not considered. Patients identified by the 40-GEP test as at highest risk of metastasis (Class 2B) consistently had metastasis rates \geq 50%, regardless of whether they had one, two or more risk factors. Similarly, the NCCN very high-risk subset of patients who were also identified as 40-GEP Class 2B had a metastatic event rate of 60.0%, although the NCCN high-risk subset with 40-GEP Class 2B results had a metastatic rate of 37.5%. While this is, indeed, a high rate of metastasis and the 40-GEP provides independent prognostication, this finding is an important reminder of the impact of baseline clinicopathologic risk on comprehensive risk assessment.

When formal tumor staging systems commonly used in the USA (AJCC8 and BWH) [21] were applied in this cohort, a similar refinement of risk stratification was demonstrated by 40-GEP class, regardless of which staging system was used. Also, statistically significant improvements were demonstrated with 40-GEP test results across several accuracy metrics. Specificity of a Class 2B (96.9%) and sensitivity of a Class 2 (77.8%) result were significantly greater than the corresponding metrics for high-stage BWH and AJCC8 tumors. Together, these metrics demonstrate the capability of the 40-GEP test to identify tumors at high risk of metastasis with improved accuracy relative to BWH and AJCC8 tumor classifications, while stratifying these cases separately from those with Class 1 tumors who have risk similar to that observed in the general cSCC patient population. Thus combining

the 40-GEP test with clinicopathologic factor-based risk assessment, regardless of whether it is based on risk factor count or T stage, can further stratify risk of metastasis in cSCC patients. These findings demonstrate the capability of molecular tools to provide enhancement of current risk assessment methods in cSCC.

While the analysis herein focuses on clinical validity of the 40-GEP assay following clinical SOPs, incorporation of the test into the guidance of patient management (i.e., clinical utility) is actively being addressed. As previously suggested by Farberg *et al.*, the 40-GEP test can be integrated within NCCN guideline recommendations to help refine risk-directed cSCC patient management decisions [51]. Also, Litchman *et al.* recently demonstrated that knowledge from the 40-GEP test can significantly impact and lead to risk-stratified change in clinician decisions for patient management relative to decisions made without 40-GEP results. Clinicians indicated they would increase management intensity with a 40-GEP Class 2B result and decrease management intensity with a Class 1 result, while remaining within national guidelines [52]. National guidelines for patient management support that the intensity of intervention/therapy/management should align with risk for poor outcomes, as in the systems proposed by Farberg *et al.* and Baum *et al.* [51,53]. As the 40-GEP predicts metastatic risk, one paradigm change could be to define the intensity of nodal assessment based on molecular prognostication, including decisions around various imaging modalities (e.g., ultrasound, MRI or CT scanning) or nodal biopsy. Indeed, the role of sentinel lymph node biopsy in management of cSCC is widely debated [54–57]. This prognostic tool could provide guidance on who might warrant the procedure because of a high likelihood of positivity; conversely, those at low risk for poor outcomes may be able to avoid this procedure.

The current study demonstrates that inclusion of molecular prognostication can augment cSCC risk stratification via objective and standardized risk classification methods. The findings from this study also show that the 40-GEP test could be helpful in predicting risk of metastasis in patients categorized as high risk by the recently updated NCCN guidelines (in which this clinical characterization alone is considered high risk for only local recurrence) [6]. Specifically, the NCCN high-risk classification is denoted by the guidelines as elevated risk of local recurrence, while the NCCN very high-risk classification is denoted as elevated risk of local recurrence and metastasis. Nevertheless, the data presented herein show that within the NCCN high-risk group there was a 15.7% risk of metastasis for Class 2A and a 37.5% risk of metastasis for Class 2B, which has potential to be clinically meaningful. Thus the significantly different risk profiles identified by the 40-GEP test on biopsy or surgical excision specimens from primary high-risk cSCC tumors could be used along with clinicopathologic factors to guide patient management decisions (i.e., follow-up frequency, nodal assessment and adjuvant therapy) [58] and, with additional studies, help define management guidelines. In addition, while efficacious treatment options for metastatic cSCC are being investigated via clinical trials and other molecular studies [58,59], the 40-GEP test could be used to stratify patients at high risk of metastasis in order to implement timely, risk-appropriate surveillance to identify and treat metastasis early. Ultimately, incorporating molecular tools into high-risk cSCC patient risk assessment could lead to more personalized and risk-appropriate methods for improved patient management and, subsequently, outcomes.

Conclusion

This study demonstrates the clinical validation and significant prognostic value of the 40-GEP test. Combining metastatic risk assessment via the clinically available 40-GEP test with clinicopathologic risk factor-based assessment can further stratify patients with cSCC and, within national guideline recommendations, could inform risk-appropriate patient management decisions.

The archival nature of samples and possible under-reporting of high-risk factors are potential limitations of this study. However, the cohort represents a high-risk cSCC population (15% metastasis rate; 60.7 and 39.3% of the cohort were NCCN high risk and very high risk, respectively) [6] and reflects current clinical pathology practices. Comprehensive monitoring of pathology reports, surgical reports and medical records, coupled with independent dermatopathologist review, was implemented to mitigate under-reporting. Another potential limitation of the study is identification of the primary lesion among multiple cutaneous lesions. However, according to the institutional review board-approved study protocol inclusion/exclusion criteria, if there was more than one cSCC in the same region, either the physician was able to unequivocally attribute the cSCC to one primary tumor (which then could be included in the study) or the tumors were excluded from the current study and were made available for future testing (reserved for synchronous tumor studies in the future). While recurrence is another cSCC tumor behavior that, like metastasis, can result in poor outcomes, the 40-GEP test was developed to classify a tumor based on risk of metastasis and at this time has not been validated for risk of local recurrence.

Future perspective

Ongoing study enrollment will allow for additional reporting for this molecular test. Currently, tissue from recurrent tumors is not being clinically tested, nor has the 40-GEP test been validated to predict risk of local recurrence, although these represent areas of ongoing research. We expect that incorporation of molecular profiling of tumors with traditional clinicopathologic factor-based assessment in patients with cSCC will evolve as a critical enhancement to patient risk assessment that better informs clinical decision-making for an optimal management strategy for each patient. This clinical enhancement for the care of patients with high-risk cSCC should lead to improved health outcomes and better use of resources.

Summary points

- The incidence of cutaneous squamous cell carcinoma (cSCC) is high (~2 million cases diagnosed/year) and continues to grow, with a substantial number of patients having poor outcomes, and an annual death toll that is likely to surpass that for melanoma.
- Most deaths from cSCC are due to progressive locoregional disease (e.g., regional lymph node metastasis), which
 underscores the clinical need for accurate identification of patients at risk of metastasis to inform
 decision-making for optimal patient management.
- While clinicopathologic factor-based risk assessment and formal staging systems are clinically helpful, they have limitations and a universally accepted system for risk stratification in cSCC has not yet been adopted.
- In a cohort (n = 420) of patients with high-risk cSCC, the intended use population, the 40-gene expression profile (40-GEP) test demonstrated significant prognostic value for stratifying metastatic risk and refined risk classification via integration of 40-GEP results with clinicopathologic factor-based assessment, identifying patients with metastasis rates near those of the general cSCC population (Class 1) and those with rates ≥50% (Class 2B).
- The 40-GEP test is clinically validated to classify risk of metastasis in patients with cSCC with one or more risk factors, and provides independent prognostic information that can enhance clinicopathologic risk factor-based assessment and established staging systems.
- Incorporating the 40-GEP test into risk assessment for patients with cSCC, within national guidelines, could facilitate better-informed decision-making for more personalized, risk-appropriate patient management and, subsequently, improved outcomes.

Supplementary data

To view the supplementary data that accompany this paper please visit the journal website at: www.futuremedicine.com/doi/sup pl/10.2217/fon-2021-1277

Author contributions

S Ibrahim, S Kurley, K Covington, S Arron and A Wysong contributed to the conception and design of the study. S Ibrahim, J Kasprzak, S Kurley, K Covington, A Farberg, S Trotter, K Reed, D Brodland, S Koyfman, A Somani, S Arron and A Wysong contributed to acquisition of data. M Hall, A Fitzgerald, J Siegel, S Kurley, K Covington and M Goldberg contributed to the analysis and interpretation of data. J Siegel and K Covington performed statistical analysis of data. S Ibrahim, M Hall, A Fitzgerald, S Kurley and A Wysong drafted the manuscript. All authors have contributed important intellectual content for critically revising the manuscript, take responsibility for integrity of the data and approve the manuscript in its final form.

Acknowledgments

The authors would like to thank the following individuals and centers for their contributions to this project: J Toyohara (Adult and Pediatric Dermatology, PC), M Gharia (Ascension/Columbia St. Mary's), C Schmults (Brigham and Women's Hospital), Y Xia (Brooke Army Medical Center), N Ezra (California Dermatology Institute), N Gharavi (Cedars Sinai), N Cleaver (Cleaver Dermatology), F Samie (Columbia University), T Schlesinger (Dermatology & Laser Center of Charleston), G Mendese (Dermatology & Skin Health), T Blalock (Emory University), J Boyer (John Boyer, MD), S Kwatra (Johns Hopkins), J Lyons (Mary Bird Perkins Cancer Center), L Toylaymat and S Fosko (Mayo Clinic, Florida), T Hansen (McFarland Clinic), D Papadopoulos (MetroDerm PC), T Knackstedt (MetroHealth, Cleveland), D Rivlin (Miami Beach Skin Center – Skin & Cancer Assoc.), N Braghiroli (Miami Cancer Institute – Baptist Health), M Fazio (Michael J. Fazio, MD, Inc.), J Fazio (Medical College of Wisconsin), J Zager (Moffitt Cancer Center), H Korasani (Mt Sinai), E Bailey (Naaman Clinic), B Brockstein (Northshore University HealthSystem), S Yoo and P Gerami (Northwestern University), C Chung (Ohio State University), A Bar (Oregon Health & Science University), R Neves (Penn State Hershey), J Campana (Porter Adventist/Centura Health Research Center), D Spearman (Premier Health – Parkview), N Fernandes (Prestige Dermatology & Medical Advisors), C Love (Radiant Complexions – Iowa Dermatology), K Brady (Roswell Park), J Canueto (Salamanca University Hospital – Spain), H

Greenway (Scripps Health), R Griego (Skin Cancer Specialists), J DeBloom (South Carolina Skin Cancer Center), M Guenther (St. Elizabeth Physicians), R Behshad (St. Louis University), M Murphy (The Indiana Skin Cancer Center), J Curry (Thomas Jefferson University Hospital), P Scumpia (UCLA Heallth), C Huang (University of Alabama), T Jennings (University of Arkansas), J Mammen (University of Kansas MC), I Maher (Saint Louis University/University of Minnesota), E Smith (University of Missouri), J Newman (University of Pennsylvania), L Ferris (University of Pittsburgh Medical Center), G Kim (University of Southern California), M Fleming (University of Tennessee Health Science Center), J Lewis and A Ward (University of Tennessee Medical Center, Knoxville), K Duffy (University of Utah), C Weinberger (University of Vermont), D Beynet (VA – Los Angeles), D Pariser (Virginia Clinical Research, Inc.), C Regula (Vujevich Dermatology Associates), J Ouellette (Wright State – Miami Valley Hospital South, OH), and F Monzon, R Cook, J Wilkinson, V Salas, L Meldi Sholl, C Bailey, K Oelschlager, C Johnson and K Carter (Castle Biosciences, Inc.).

Financial & competing interests disclosure

Materials and funding for this study were provided by Castle Biosciences, Inc. S Ibrahim receives research funding from Regeneron and funding for lectures/speakers bureau services from Regeneron and Genentech; M Hall, A Fitzgerald, J Siegel, S Kurley, K Covington and M Goldberg are employees and options holders for Castle Biosciences, Inc.; A Farberg serves as a consultant for Castle Biosciences, Inc. and is on the advisory boards for Eli Lilly, SunPharma, Orthodermatologics, Boehringer Ingelheim, Incyte, Amgen, Galderma, Novartis, and Pfizer; S Koyfman receives research funding and consulting fees from Merck, research funding from Bristol Myers Squibb, and honoraria from UpToDate; S Arron has been an employee of Rakuten Medical and is currently a consultant for Castle Biosciences, Inc. and Enspectra Health; and A Wysong serves on the WDS and ACMS Boards of Directors and is a Trustee for the Dermatology Foundation (nonmonetary service) and Castle Biosciences, Inc. Steering Committee (nonmonetary service). All remaining authors participated as investigators for Castle Biosciences, Inc. during this study. The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

No writing assistance was utilized in the production of this manuscript.

Ethical conduct of research

The authors have obtained appropriate institutional review board (IRB) approval via Western IRB and informed consent has been obtained from human study participants as required.

Open access

This work is licensed under the Attribution-NonCommercial-NoDerivatives 4.0 Unported License. To view a copy of this license, visit http://creativecommons.org/licenses/by-nc-nd/4.0/

References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- 1. Brantsch KD, Meisner C, Schönfisch B *et al.* Analysis of risk factors determining prognosis of cutaneous squamous-cell carcinoma: a prospective study. *Lancet Oncol.* 9(8), 713–720 (2008).
- 2. Belkin D, Carucci JA. Mohs surgery for squamous cell carcinoma. Dermatol. Clin. 29(2), 161-174 (2011).
- Brougham N, Dennett E, Cameron R, Tan S. The incidence of metastasis from cutaneous squamous cell carcinoma and the impact of its risk factors. J. Surg. Oncol. 106, 811–815 (2012).
- Karia PS, Morgan FC, Ruiz ES, Schmults CD. Clinical and incidental perineural invasion of cutaneous squamous cell carcinoma. JAMA Dermatol. 153(8), 781 (2017).
- 5. Que SKT, Zwald FO, Schmults CD. Cutaneous squamous cell carcinoma: incidence, risk factors, diagnosis, and staging. J. Am. Acad. Dermatol. 78(2), 237–247 (2018).
- Study describes estimates of incidence, risk factors and staging.
- 6. National Comprehensive Cancer Network. Squamous cell skin cancer, NCCN Guidelines version 2.2021 (2021). www.nccn.org/professionals/physician_gls/pdf/squamous.pdf
- The current national guidelines for risk assessment and patient management for cutaneous squamous cell carcinoma (cSCC).
- 7. Karia PS, Han J, Schmults CD. Cutaneous squamous cell carcinoma: estimated incidence of disease, nodal metastasis, and deaths from disease in the United States, 2012. *J. Am. Acad. Dermatol.* 68(6), 957–966 (2013).
- 8. Waldman A, Schmults C. Cutaneous squamous cell carcinoma. Hematol. Oncol. Clin. North Am. 33(1), 1-12 (2019).
- 9. Rogers HW, Weinstock MA, Feldman SR, Coldiron BM. Incidence estimate of nonmelanoma skin cancer (keratinocyte carcinomas) in the US population, 2012. *JAMA Dermatol.* 151(10), 1081 (2015).
- 10. Skin Cancer Foundation. Skin cancer facts & statistics (2021). www.skincancer.org/skin-cancer-information/skin-cancer-facts/

- 11. Skin Cancer Foundation. Our new approach to a challenging skin cancer statistic (2021). www.skincancer.org/blog/our-new-approach-to-a-challenging-skin-cancer-statistic/
- 12. High-Risk Cutaneous Squamous Cell Carcinoma: A Practical Guide for Patient Management. Schmults CD. (Ed.). Springer, Berlin/Heidelberg, Germany (2016).
- 13. Schmults CD, Karia PS, Carter JB, Han J, Qureshi AA. Factors predictive of recurrence and death from cutaneous squamous cell carcinoma: a 10-year, single-institution cohort study. *JAMA Dermatol.* 149(5), 541 (2013).
- 14. Harris BN, Pipkorn P, Nguyen KNB et al. Association of adjuvant radiation therapy with survival in patients with advanced cutaneous squamous cell carcinoma of the head and neck. JAMA Otolaryngol. Head Neck Surg. 145(2), 153 (2019).
- 15. Migden MR, Khushalani NI, Chang ALS et al. Cemiplimab in locally advanced cutaneous squamous cell carcinoma: results from an open-label, Phase 2, single-arm trial. Lancet Oncol. 21(2), 294–305 (2020).
- Thompson AK, Kelley BF, Prokop LJ, Murad MH, Baum CL. Risk factors for cutaneous squamous cell carcinoma outcomes: a systematic review and meta-analysis. *JAMA Dermatol.* 152(4), 419–428 (2016).
- 17. Alam M, Ratner D. Cutaneous squamous-cell carcinoma. N. Engl. J. Med. 344(13), 975-983 (2001).
- Alam M, Armstrong A, Baum C et al. Guidelines of care for the management of cutaneous squamous cell carcinoma. J. Am. Acad. Dermatol. 78(3), 560–578 (2018).
- Study provides guidelines for cSCC patient management.
- American Joint Commission on Cancer. Amin MB, Edge S, Greene F et al. (Eds). AJCC Cancer Staging Manual. 8th Edition. Springer International Publishing, NY, USA (2017).
- This is the current edition of the American Joint Committee on Cancer Cancer Staging Manual (AJCC) Cancer Staging Manual.
- 20. Ruiz ES, Karia PS, Besaw R, Schmults CD. Performance of the American Joint Committee on Cancer Staging Manual, 8th Edition vs the Brigham and Women's Hospital Tumor Classification System for cutaneous squamous cell carcinoma. *JAMA Dermatol.* 155(7), 819 (2019).
- Study describes performance (including accuracy metrics) of AJCC8 and Brigham and Women's Hospital (BWH) systems.
- Karia PS, Jambusaria-Pahlajani A, Harrington DP, Murphy GF, Qureshi AA, Schmults CD. Evaluation of American Joint Committee on Cancer, International Union Against Cancer, and Brigham and Women's Hospital tumor staging for cutaneous squamous cell carcinoma. JCO 32(4), 327–334 (2014).
- 22. Karia PS, Morgan FC, Califano JA, Schmults CD. Comparison of tumor classifications for cutaneous squamous cell carcinoma of the head and neck in the 7th vs 8th Edition of the AJCC Cancer Staging Manual. *JAMA Dermatol.* 154(2), 175 (2018).
- 23. Wysong A, Newman JG, Covington KR et al. Validation of a 40-gene expression profile test to predict metastatic risk in localized high-risk cutaneous squamous cell carcinoma. J. Am. Acad. Dermatol. 84(2), 361–369 (2021).
- •• Study describes development and validation of the algorithm for the 40-gene expression profile (40-GEP).
- 24. Jambusaria-Pahlajani A, Kanetsky PA, Karia PS *et al.* Evaluation of AJCC tumor staging for cutaneous squamous cell carcinoma and a proposed alternative tumor staging system. *JAMA Dermatol.* 149(4), 402 (2013).
- Chu MB, Slutsky JB, Dhandha MM *et al.* Evaluation of the definitions of 'high-risk' cutaneous squamous cell carcinoma using the American Joint Committee on Cancer Staging Criteria and National Comprehensive Cancer Network Guidelines. *J. Skin Cancer* 2014(4), 1–8 (2014).
- 26. DeSimone JA, Karia PS, Hong AM, Ruiz ES, Jambusaria-Pahlajani A. Recognition and management of high-risk (aggressive) cutaneous squamous cell carcinoma. In: *Up ToDate*. Corona R, Robinson JK (Eds). UpToDate, MA, USA (2020).
- Tschetter AJ, Campoli MR, Zitelli JA, Brodland DG. Long-term clinical outcomes of patients with invasive cutaneous squamous cell carcinoma treated with Mohs micrographic surgery: a 5-year, multicenter, prospective cohort study. J. Am. Acad. Dermatol. 82(1), 139–148 (2020).
- Marrazzo G, Zitelli JA, Brodland D. Clinical outcomes in high-risk squamous cell carcinoma patients treated with Mohs micrographic surgery alone. J. Am. Acad. Dermatol. 80(3), 633–638 (2019).
- 29. Haisma MS, Plaat BEC, Bijl HP *et al.* Multivariate analysis of potential risk factors for lymph node metastasis in patients with cutaneous squamous cell carcinoma of the head and neck. *J. Am. Acad. Dermatol.* 75(4), 722–730 (2016).
- 30. Cañueto J, Burguillo J, Moyano-Bueno D et al. Comparing the eighth and the seventh editions of the American Joint Committee on Cancer staging system and the Brigham and Women's Hospital alternative staging system for cutaneous squamous cell carcinoma: implications for clinical practice. J. Am. Acad. Dermatol. 80(1), 106–113.e2 (2019).
- 31. Teplitz R, Giselle P, Litchman GH, Rigel DS. Impact of gene expression profile testing on the management of squamous cell carcinoma by dermatologists. *J. Drugs Dermatol.* 18(10), 980–984 (2019).
- 32. Conde-Ferreirós A, Corchete LA, Puebla-Tornero L *et al.* Definition of prognostic subgroups in the T3 stage of the eighth edition of the American Joint Committee on Cancer staging system for cutaneous squamous cell carcinoma: tentative T3 stage subclassification. *J. Am. Acad. Dermatol.* 85(5), 1168–1177 (2020).

- 33. National Comprehensive Cancer Network. Squamous Cell Skin Cancer, NCCN Guidelines Version 2.2020 (2020). www.nccn.org/professionals/physician_gls/default.aspx#site
- 34. Vidal CI, Armbrect EA, Andea AA *et al.* Appropriate use criteria in dermatopathology: initial recommendations from the American Society of Dermatopathology. *J. Cutan. Pathol.* 45(8), 563–580 (2018).
- 35. Rubin AI, Hitchcock M. A first for dermatopathology and pathology in general: appropriate use criteria in dermatopathology from the American Society of Dermatopathology. J. Cutan. Pathol. 45(8), 561–562 (2018).
- Genders RE, Weijns ME, Dekkers OM, Plasmeijer EI. Metastasis of cutaneous squamous cell carcinoma in organ transplant recipients and the immunocompetent population: is there a difference? A systematic review and meta-analysis. J. Eur. Acad. Dermatol. Venereol. 33(5), 828–841 (2019).
- 37. Yildiz P, Aung PP, Milton DR *et al.* Measurement of tumor thickness in cutaneous squamous cell carcinomas: do the different methods provide better prognostic data? *Am. J. Dermatopathol.* 42(5), 337–342 (2020).
- 38. Alford AV, Brito JM, Yadav KK, Yadav SS, Tewari AK, Renzulli J. The use of biomarkers in prostate cancer screening and treatment. *Rev. Urol.* 19(4), 221–234 (2017).
- 39. Scope A, Essat M, Pandor A et al. Gene expression profiling and expanded immunohistochemistry tests to guide selection of chemotherapy regimens in breast cancer management: a systematic review. Int. J. Technol. Assess. Health Care 33(1), 32–45 (2017).
- 40. Vetto JT, Hsueh EC, Gastman BR *et al.* Guidance of sentinel lymph node biopsy decisions in patients with T1–T2 melanoma using gene expression profiling. *Future Oncol.* 15(11), 1207–1217 (2019).
- 41. Vargas-Salas S, Martínez JR, Urra S *et al.* Genetic testing for indeterminate thyroid cytology: review and meta-analysis. *Endocr. Relat. Cancer* 25(3), R163–R177 (2018).
- 42. Plasseraud KM, Wilkinson JK, Oelschlager KM *et al.* Gene expression profiling in uveal melanoma: technical reliability and correlation of molecular class with pathologic characteristics. *Diagn. Pathol.* 12(1), 59 (2017).
- Onken MD, Worley LA, Char DH et al. Collaborative Ocular Oncology Group Report Number 1: prospective validation of a multi-gene prognostic assay in uveal melanoma. *Ophthalmology* 119(8), 1596–1603 (2012).
- 44. Hsueh EC, DeBloom JR, Lee J *et al.* Interim analysis of survival in a prospective, multi-center registry cohort of cutaneous melanoma tested with a prognostic 31-gene expression profile test. *J. Hematol. Oncol.* 10(1), 152 (2017).
- Maly CJ, Cumsky HJL, Costello CM et al. Prognostic value of inositol polyphosphate-5-phosphatase expression in recurrent and metastatic cutaneous squamous cell carcinoma. J. Am. Acad. Dermatol. 82(4), 846–853 (2019).
- Lobl M, Grinnell M, Phillips A, Abels J, Wysong A. The correlation between immunohistochemistry findings and metastasis in squamous cell carcinoma: a review. *Dermatol. Surg.* 47(3), 313–318 (2020).
- Mitsui H, Suárez-Fariñas M, Gulati N et al. Gene expression profiling of the leading edge of cutaneous squamous cell carcinoma: IL-24-driven MMP-7. J. Invest. Dermatol. 134(5), 1418–1427 (2014).
- 48. Choi KH, Kim GM, Kim SY. The keratin-14 expression in actinic keratosis and squamous cell carcinoma: is this a prognostic factor for tumor progression? *Cancer Res. Treat.* 42(2), 107–114 (2010).
- 49. Chitsazzadeh V, Coarfa C, Drummond JA *et al.* Cross-species identification of genomic drivers of squamous cell carcinoma development across preneoplastic intermediates. *Nat. Commun.* 7, 12601 (2016).
- 50. McLaughlin EJ, Miller L, Shin TM *et al.* Rate of regional nodal metastases of cutaneous squamous cell carcinoma in the immunosuppressed patient. *Am. J. Otolaryngol.* 38(3), 325–328 (2017).
- 51. Farberg AS, Hall MA, Douglas L *et al.* Integrating gene expression profiling into NCCN high-risk cutaneous squamous cell carcinoma management recommendations: impact on patient management. *Curr. Med. Res. Opin.* 36(8), 1301–1307 (2020).
- Study proposed patient management strategies for incorporation of 40-GEP testing into clinical management decisions for high risk cSCC.
- 52. Litchman GH, Fitzgerald AL, Kurley SJ, Cook RW, Rigel DS. Impact of a prognostic 40-gene expression profiling test on clinical management decisions for high-risk cutaneous squamous cell carcinoma. *Curr. Med. Res. Opin.* 36(8), 1295–1300 (2020).
- 53. Baum CL, Wright AC, Martinez J-C *et al.* A new evidence-based risk stratification system for cutaneous squamous cell carcinoma into low, intermediate, and high risk groups with implications for management. *J. Am. Acad. Dermatol.* 78(1), 141–147 (2018).
- 54. Allen JE, Stolle LB. Utility of sentinel node biopsy in patients with high-risk cutaneous squamous cell carcinoma. *Eur. J. Surg. Oncol.* 41(2), 197–200 (2015).
- 55. Navarrete-Dechent C, Veness MJ, Droppelmann N, Uribe P. Cutaneous squamous cell carcinoma and the emerging role of sentinel lymph node biopsy. *G. Ital. Dermatol. Venereol.* 153(3), 403–418 (2018).
- 56. Wu MP, Sethi RKV, Emerick KS. Sentinel lymph node biopsy for high-risk cutaneous squamous cell carcinoma of the head and neck. *Laryngoscope* 130(1), 108–114 (2020).
- 57. Tremblay-Abel V, Poulin M-A, Blouin M-M, Parent F, Perron É. Sentinel lymph node biopsy in high-risk cutaneous squamous cell carcinoma: analysis of a large size retrospective series. *Dermatol. Surg.* 47(7), 908–913 (2021).

- 58. Newman JG, Hall MA, Kurley SJ et al. Adjuvant therapy for high-risk cutaneous squamous cell carcinoma: a 10-year review. Head Neck 43(9), 2822–2843 (2021).
- 59. Ribero S, Stucci LS, Daniels GA, Borradori L. Drug therapy of advanced cutaneous squamous cell carcinoma: is there any evidence? *Curr. Opin. Oncol.* 29(2), 129–135 (2017).